Author's response to reviews

Title: The role of Traditional Birth Attendants in urban and rural Tanzania: a mixed-methods study

Authors:

Constanze D Pfeiffer (constanze.pfeiffer@unibas.ch)
Rosemarie Mwaipopo (ny_lila@yahoo.com)

Version: 5 Date: 14 December 2012

Author's response to reviews: see over